MCC Operations Status Meeting

February 4, 2008
This afternoons 4:15pm meeting:
Everyone:
Please submit your mini-test plan for Wednesday’s Software Day
Feedback: Diane Fairley

Potential timing issue when reading power meters before wave plates have reached desired set point. Feedback software may need to have delays added. See Joe Frisch for details. 

LLRF: High voltage output drifting, Ron Aker

Profile Monitor:  Sheng Peng and Zen Szalata

The test camera in the gallery gets its trigger from a signal generator. Since no problems have been seen, Sheng will continue testing but will use triggers from the evr for this camera.

Wire Scanner s: Arturo Alarcon
1)After the meeting will want 40 minute to implement changes to resolve issues listed in the morning meeting.

2) Soft limits were reached in a wire scanner but were unable to reset this do to db limits. See Jim Turner for more information.

3) Beam was seen on wire scanners, so wires are ok and problems listed this morning point to software issue.

This mornings 8:00am meeting:

Everyone:  Not all Artemis entries  listed below, please check.

Aida: Greg White

Entering PDES for rf from matlab causes a trim. However, when PDES is entered from a SCP the trim does not occur.
Computer Systems:  Jingchen Zhou

1) Artemis #82603. Right arrow key on EIOC desk linux box (on all linux boxes in mcc), has been remapped as backspace/delete key.
2) Artemis #82594. Elog problem displaying images. Problem is intermittent. E-log graphics turned into blue boxes on many pages (unreadable)

3) Artemis #82593. The scp software knobs display brought up on an opi linux box uses a huge font and the display is unusable.
Laser: Sheng Peng and Zen Szalata

Laser energy change observed upon repetition rate change from 10 to 30 Hz, or the reverse.

LLRF: Dayle Kotturi and Ron Akre

New PACs in 24-1 & 24-2 allow a BC2 energy feedback loop - a big improvement.

Magnets:  Kristi  Luchini

1. There is no archive history data for BX12 an LI24  PSOUT
2. LGPS LI22-24 tripped, water flow problem. PEM is looking into this morning. 

Miscellaneous:   Fixed.
1) Artemis #82595 – Problems with operator instructions for laser. Sasha has corrected instructions.

Profile Monitors: Sheng Peng and Zen Szalata

Needed to reboot OTR12 twice, but never saw any video problem on Sheng's "Tester"
 video. The problem seems to be tunnel issue and not timing.

SCORE:  Debbie Rogind

1. Artemis #82597.  User prompted about injector when saving bsy. 

Toroids:  Sheng Peng, Zen Szalata and Mike Cecere

Toroid calibration looks bad. See Elog from 2/03/2008 20:36.  Toroids have been bypassed so they can run machine

VMS Systems:  Nancy Spencer

LCLS and PEP Klystron CUDS  intermittently non-operational.  A similar Artemis entry was closed but the problem is not fixed.  No error message given, the CUDs just stop working.
Wire Scanners: Arturo Alarcon

1) Artemis #82599. Problems with WS12 over the weekend.  However WS02 an WS01 also had problems. Suspect scan distances is small and calculated speed is too low. 

2) Setting wire-scan ranges is very tricky and time consuming (plot scale and range scales do
 not agree).

3) Frequent inconsequential wire scan errors reported (something like: 'Wire scan failed',
 but clicking on "proceed" always works with no obvious errors - is this Doug's
 2-micron LVDT-motor error?).

4) Cryptic WS12 error reported: "No wire position in desired scan range!  Problem in EPICS!
  Call x5346."  (3 options available: "CRASH", "CRASH", or "CRASH").
5) Wire scanners completely failed - no ion chamber data or wire position data reported at all
 (WS12 primarily)

6) LCLS phyisicist would like to archive Gate Delay, Gate Width, and Requested Pedestal for all PMTs.
